Author’s response to reviews

Title: Written narratives from immigrants following a prenatal diagnosis: qualitative exploratory study

Authors:

Tommy Carlsson (tommy.carlsson@kbh.uu.se)
Banaz Balbas (banaz7@live.se)
Elisabet Mattsson (elisabet.mattsson@esh.se)

Version: 3 Date: 29 Mar 2019

Author’s response to reviews:

Dear Editor and Reviewers,

Thank you for giving us the opportunity to revise the manuscript entitled “Written narratives from immigrants following a prenatal diagnosis: qualitative exploratory study”, submitted to BMC Pregnancy and Childbirth (PRCH-D-18-00529). Please see responses to specific comments below.

EDITORS COMMENTS

1. Please clarify the consent for publication statement, by indicating when participants consented for publications of their quotes.

We have added the following information: ” The respondents gave their consent for direct quotes to be published when sending in their responses.”

2. Please clarify whether the ethics committee approved the consent procedures used in your study, or whether national guidelines dictate that the form of consent used is OK for studies such as this.

We have added the following information: ” The study was approved by the Regional Ethical Committee in Uppsala, Sweden (approval number 2014/504). The committee approved the consent procedure.”
3. Please upload English language versions of all questionnaires developed specifically for use in this study as supplementary files. If the questionnaires are not yours or are already published by you, please supply references and/or links to them.

We have added an additional file with the questionnaire.

3. As a reminder, upon uploading your revisions, please remove any tracked changes or highlighting and include only a single clean copy of the manuscript.

We have uploaded a clean copy of the manuscript.

With kind regards

On behalf of all authors

Tommy Carlsson